Author's response to reviews

Title: Good functional recovery following intervention for delayed suprachoroidal haemorrhage post bleb needling: a case report

Authors:

Paul S Cannon (pscan05@yahoo.co.uk)
Fiona A Spencer (fiona.spencer@CMMC.nhs.uk)
Michael J Lavin (lavin@CMMC.nhs.uk)

Version: 9 Date: 4 March 2008

Author's response to reviews: see over
Dear Editorial Team,

Please find the following changes as requested.

1. The author list is identical in both the manuscript and submission system.

2. References - References should be cited in the text using consecutive numbers in brackets.

3. Consent - This has been altered as follows “Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.”

4. References- The reference list has been formatted as you requested.

5. Figure cropping – as requested.

6. Please supply a higher resolution copy of figure 2.

Yours truly,

Paul S. Cannon
MRCOphth, MRCPI